Author's response to reviews

Title: Effect of physical activity intervention based on a pedometer on physical activity level and anthropometric measures after childbirth: a randomized controlled trial

Authors:

Masumeh SechinMaturi (nastaran47@yahoo.com)
Pourandokht Afshary (p_afshary@yahoo.com)
Parvin Abedi (parvinabedi@ymail.com)

Version: 6 Date: 7 December 2011

Author's response to reviews:

I made all changes you have requested.